Author's response to reviews

Title: Effect of religiosity/spirituality and sense of coherence on depression within a rural population in Greece: the Spili III project

Authors:

Dimitrios Anyfantakis (danyfantakis@med.uoc.gr)
Emmanouil K Symvoulakis (symvouman@yahoo.com)
Manolis Linardakis (linman@med.uoc.gr)
Sue Shea (sueshea1@otenet.gr)
Demosthenes Panagiotakos (d.b.panagiotakos@usa.net)
Christos Lionis (lionis@galinos.med.uoc.gr)

Version: 2 Date: 15 January 2015

Author's response to reviews:

Dear Editor,

Many thanks for your message and we revised the manuscript following your instructions. More specifically:

1. The title page has been corrected, email addresses have been included and a full postal address of the submitting author.
2. Line numbering has been added in the manuscript.
3. An acknowledgements section has been included prior to the Reference list.

On behalf of all co-authors,

Christos Lionis